Author's response to reviews

Title: Clients of sex workers in Switzerland: it makes sense to counsel and propose rapid test for HIV on the street.

Authors:

Esther-Amélie Diserens (esther_amelie@hospvd.ch)
Patrick Bodenmann (patrick.bodenmann@hospvd.ch)
Chantal N'Garambe (chantal.ngarambe@hospvd.ch)
Anne Ansermet-Pagot (fleurdepave@bluewin.ch)
Marco Vannotti (marco.vannotti@hospvd.ch)
Eric Masserey (eric.masserey@vd.ch)
Matthias Cavassini (matthias.cavassini@chuv.ch)

Version: 4 Date: 14 March 2010

Author's response to reviews: see over
Dear Editorial Production Team,

We thank you for your comments. Please find below the point-by-point description of the changes made.

1. **Title:** manuscript title is the correct one, we have modified the one in the submission system

2. **Author list:** list of author has been modified and written with the full first name as in the submission system

3. **Affiliations:** each affiliation is now written in full with Department, Institution, City, Country. We did not write the (Street) nor the (postal code). If you want these informations, please contact me again by e-mail.

4. **Page 14 (blank) has been removed (after the figure)**

5. **Figure:** we deleted the title

6. **Typography:** we checked again for typographical errors.

Best regards,

On Behalf of the authors,

M. Cavassini

Matthias.cavassini@chuv.ch